Author's response to reviews

Title: Post-Bronchoscopy Fatal Endo-Bronchial Hemorrhage in Broncho-Pulmonary Mucormycosis: a case report

Authors:

Paola P Di Carlo (pdicarlo@unipa.it)
Daniela D Cabibi (cabibi@unipa.it)
Anton Maria AM La Rocca (antonmarialarocca@yahoo.it)
Dario D De Luca (dariodeluca@virgilio.it)
Francesco F La Licata (f.lalicata@libero.it)
Ennio E Sacco (saccoennio@libero.it)

Version: 3 Date: 22 June 2010

Author's response to reviews: see over
To The JMCR Editorial Team,

Please find herewith included an electronic version of the revised version of the manuscript entitled: ““Post-Bronchoscopy Fatal Endo-Bronchial Hemorrhage in Broncho-Pulmonary Mucormycosis: a case report”. It has been assigned tracking no. MS: 9157246413405544.

Authors: Paola Di Carlo, Daniela Cabibi, Anton Maria La Rocca, Dario De Luca, Francesco La Licata and Ennio Sacco.

The Authors thank the Associate Editor for providing constructive comments and help in improving the contents of the paper.

As explained in detail below, the authors have made a number of changes, taking into account these constructive remarks.

<<he feels that it is still rather disjointed which makes it difficult to read. Also, there are few errors in bibliography and legends sections. He has requested further revision giving special attention to the grammar and coherence >>

In order to take into account the Editor’s important suggestions, and in accordance with the Journal’s guidelines on setting out a case report, the authors have modified:

- the title: ““Post-Bronchoscopy Fatal Endo-Bronchial Hemorrhage in Broncho-Pulmonary Mucormycosis: a case report”;
- the order of the comments in the Discussion for a more coherent reading. The changes have been highlighted;
- the conclusions, in an attempt to present them in a more concise manner;
- the English, which has again been revised and corrected by a mother-tongue speaker;
- the figure legends, following the authors’ guidelines;
- figure 2, which now has a black background for clearer reading;
- the references, reducing their number as indicated in the authors’ guidelines, and a more recent reference has been added;
- the bibliography section layout, as indicated in the authors’ guidelines.
Sincerely,
Paola Di Carlo, MD

**Corresponding author**

Dr. Paola Di Carlo, Infectious Diseases Section, Department of Health Promotion Sciences, University of Palermo, Via del Vespro 129, 90127 Palermo, Italy (pdicarlo@unipa.it)